Author’s response to reviews

Title: Methodological considerations for a randomised controlled trial of podiatry care in rheumatoid arthritis: lessons from an exploratory trial

Authors:

Deborah E Turner (debbie.turner@gcal.ac.uk)
Philip S Helliwell (p.helliwell@leeds.ac.uk)
James Woodburn (jim.woodburn@gcal.ac.uk)

Version: 4 Date: 5 November 2007

Author’s response to reviews:

Thank you for the opportunity to submit a revised version of this manuscript. The following format changes have been made:

1- the text 'By' and 'Address' have been removed from the title page.
2- At competing interests 'author(s)' has been changed to 'authors'.
3- Ref 1 has now been published and the full details are provided.
4- Ref 18 contains the names of all the authors for this paper.
5- The tables have been moved to text and formatted as recommended.